Author's response to reviews

Title: Artemisia annua L.: Sesquiterpene lactone fraction antinociceptive activity

Authors:

Fabricio de Faveri Favero (fabricioffavero@gmail.com)
Rogério Grando (roggrando2@yahoo.com.br)
Fabiana R Nonato (frnonato@uol.com.br)
Giovanna B Longato (giovannabl@yahoo.com.br)
Ilza MO Sousa (ilza@cpqba.unicamp.br)
Núbia CA Queiroz (nubia.cassia@hotmail.com)
João E Carvalho (carvalho_je@yahoo.com.br)
Humberto M Spindola (hmspindola@bol.com.br)
Mary A Foglio (foglioma@gmail.com)

Version: 3
Date: 6 March 2014

Author's response to reviews: see over
Cover Letter

Dear Sir,

We apologize for the mistake. We have a standard letter that we use when submitting manuscripts and were unaware another journal’s name was written.

We here declare that the manuscript is not currently under consideration at any journal other than BMC Complementary and Alternative Medicine.

Hoping the data presented herein meet the high standards of BMC Complementary and Alternative Medicine.

Best regards,

Mary Ann Foglio